%% Alert: Much stuff is commented out with \old

\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P31}{20(2)}{2013}

\usepackage{amsthm,amsmath,amssymb}

\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

%\usepackage{amsmath,paralist}
%\usepackage{amsthm}
\usepackage{graphicx}
%\usepackage{tikz,color}
%\usepackage[margin=1in]{geometry}
%\usetikzlibrary{arrows,shapes,snakes,backgrounds} 
%\usepackage{epsfig}
%\usepackage{graphics}

%\usepackage{bbm,amssymb}
%\usepackage{bbold}
%\usepackage{graphicx}
%\usepackage{hyperref}
 
%\def\topfraction{.9}
%\def\floatpagefraction{.8}

\theoremstyle{plain}

%\newtheorem{theorem}{Theorem}[Section]
\newtheorem{theorem}{Theorem}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{conjecture}[theorem]{Conjecture}
 
\theoremstyle{remark}
\newtheorem*{remark}{Remark}
\newtheorem*{example}{Example}

\theoremstyle{definition}
\newtheorem*{definition}{Definition}
 

%\newcommand{\com}[1]{\textcolor{red}{$[\star$ #1 $\star]$}}


  \def\e{\epsilon}
   \def\vol{{\rm vol}}
\def\ca{{\rm Cox^A(n)}}
\def\cb{{\rm Cox^{C}(n)}}
\def\cc{{\rm Cox^{C}(n)}}

\def\x{{\bf x}}
\def\l{{\lambda}}
 
\def\c{{\bf c}}
\def\o{{\bf o }}
\def\fin{{\rm fin}}
\def\v{{\bf a}}
\def\a{{\alpha}}
 \def\f_H{{\bf w}}
 \def\b{{\bf b}}
 \def\a{{\bf a}}
\def\R{\mathbb{R}}
\def\N{\mathbb{N}}
\def\Z{\mathbb{Z}}
\def\aa{{\bf a}}
 \def\A{\mathcal{A}}
 \def\H{\mathcal{H}}
 \def\S{\mathfrak{S}}
 \def\sa{\mathcal{S}^A_n}
  \def\sb{\mathcal{S}^B_n}
   \def\scc{\mathcal{S}^C_n}
\def\san{{{(\mathcal{S}^A_n)}^{\sqcup (n+1)}}}

\newcommand{\old}[1]{}

 


\title{Labeling the regions of the type $C_n$ Shi arrangement}

\author{Karola M\'esz\'aros\thanks{The author is partially supported by a National Science Foundation Postdoctoral Research Fellowship (DMS 1103933).}\\
\small Department of Mathematics\\[-0.8ex]
\small Cornell University\\[-0.8ex] 
\small Ithaca, NY 14853
%\small\tt remifa@dis.solatido.edu\\
}

\date{\dateline{Jan 24, 2012}{May 9, 2013}{May 16, 2013}\\
\small Mathematics Subject Classifications: 52C35, 06A07}

%\keywords{type  $C_n$ Shi arrangements, sequences, posets,  nonnesting partitions.}
%\subjclass[2010]{52C35, 06A07}

\begin{document}
\begin{thebibliography}{99}
\bibitem{arm} {\sc D.~Armstrong},   \newblock Generalized noncrossing partitions and combinatorics of Coxeter groups,   \newblock \textit{Memoirs of the American Mathematical Society}, (2009), No. 949. 
\bibitem{arm-rei-rho} {\sc D.~Armstrong, V. ~Reiner, B.~Rhoades},   \newblock Parking spaces,   \newblock manuscript, 2011.
%\bibitem[3]{arm-rho} {\sc D.~Armstrong, B.~Rhoades}, The Shi Arrangement and the Ish Arrangement, \textit{to appear in Transactions of the American Mathematical Society}.  
\bibitem{ath} {\sc C.~Athanasiadis},   \newblock On noncrossing and nonnesting partitions for classical reflection groups,   \newblock \textit{Electronic J. Combin.} \textbf{5} (1998), R42. 
\bibitem{ath-lin} {\sc C.~Athanasiadis, S.~Linusson},   \newblock A simple bijection for the regions of the Shi arrangment of hyperplanes,    \newblock \textit{Discrete Math.} \textbf{204} (1999),  27-39.  
\bibitem{cel-pap} {\sc P.~Cellini, P.~Papi},   \newblock ad-nilpotent ideals of a Borel subalgebra II,   \newblock \textit{J. Algebra} \textbf{258} (2002), 112-121.    
\bibitem{fin-iri} {\sc A.~Fink, B.~Iriarte},   \newblock Bijections between noncrossing and nonnesting partitions for classical reflection groups,    \newblock \textit{Portugaliae Mathematicae} \textbf{67} fasc. 3 (2010), 369-401.
%\bibitem[8]{kre} {\sc G.~Kreweras}, Sur les partitions non-crois\'{e}es d'un cycle, \textit{Discrete Math.} \textbf{1} (1972), 333-350. 
\bibitem{shi} {\sc J.-Y.~Shi},   \newblock The Kazhdan-Lusztig cells in certain affine Weyl groups,   \newblock \textit{Lecture Notes in Mathematics}, No. 1179, Springer, Berlin, 1986. 
\bibitem{shi2} {\sc J.-Y.~Shi},   \newblock Sign types corresponding to an affine Weyl group,   \newblock \textit{J. London Math. Soc.} \textbf{35} (1987), 56-74.
\bibitem{sta0} {\sc R.~Stanley},   \newblock Hyperplane arragments, interval orders, and trees,   \newblock \textit{Proc. Nat. Acad. Sci.}, \textbf{93} (1996), 2620-2625. 
\bibitem{sta} {\sc R.~Stanley},   \newblock Hyperplane arrangements, parking functions and tree inversions,   \newblock \textit{Mathematical Essays in Honor of Gian-Carlo Rota} (B. Sagan and R. Stanley, eds.), Birkh\"{a}user, Boston/Basel/Berlin, 1998, pp. 259-375. 
\bibitem{ec1} {\sc R.~Stanley},   \newblock \emph{Enumerative Combinatorics}, vol.\ 1, 
  \newblock Cambridge University Press, New York/Cambridge, 1997.
 
%\bibitem[14]{ec2}  {\sc R. Stanley}, \emph{Enumerative Combinatorics}, vol.\ 2, 
%Cambridge University Press, New York/Cambridge, 1999.


\end{thebibliography}


\maketitle

\begin{abstract} The number of  regions of the  type $C_n$  Shi arrangement  in $\R^n$ is $(2n+1)^n$. Strikingly, no  bijective proof of this fact has been given thus far.  The aim of this paper is to provide such a bijection and use it to prove more refined results.  We construct a bijection between the regions of the  type $C_n$  Shi arrangement  in $\R^n$ and sequences $a_1a_2 \ldots a_n$, where $a_i \in \{-n, -n+1, \ldots, -1, 0, 1, \ldots, n-1, n\}$, $ i \in [n]$.  
Our bijection naturally restrict to bijections between special regions  of the arrangement and sequences with a given number of distinct elements.  
  
  \bigskip\noindent \textbf{Keywords:} type  $C_n$ Shi arrangements, sequences, posets,  nonnesting partitions.
  \end{abstract}

 
\maketitle
  
  \section{Introduction}
  \label{sec:intro}
    
    A \textbf{hyperplane arrangement}  $\A$ is a finite set of affine hyperplanes in $\R^n$. The \textbf{regions} of $\A$ are the connected components of the space $\R^n \backslash \cup_{H \in \A} H$. 
       In this paper we  study the \textbf{type $C_n$ Shi arrangement}, which is an affine  hyperplane arrangement whose hyperplanes are parallel to reflecting hyperplanes of the type $C_n$  Coxeter group.  The closely related  type $A_{n-1}$ Shi arrangement  has been much studied before. 
      \old{Recall that  ${\rm Cox}^A(n)$ the \textbf{Coxeter arrangement of type $A_{n-1}$:}
   $${\rm Cox}^A(n)=\{x_i-x_j=0 \mid 1\leq i<j\leq n\}.$$
    
    Note that the regions of  ${\rm Cox}^A(n)$ are naturally indexed by type $A_{n-1}$ permutations $w \in \mathfrak{S}_n$. Namely, if $C^A\subset \R^n$ is the \textbf{dominant cone} of  ${\rm Cox}^A(n)$  defined by $x_1>x_2>\cdots>x_n$, then 
    $$ wC^A=\{ \x \in \R^n | x_{w(1)}>x_{w(2)}>\cdots >x_{w(n)} \}.$$ 
    
    Thus, the number of regions of $\ca$ is $|\mathfrak{S}_n|=n!.$ The \textbf{type $A_{n-1}$ Shi arrangement $\sa$} was first defined by Shi \cite{shi}:
    $$\sa=\ca \cup \{x_i-x_j=1 \mid 1\leq i<j\leq n\}.$$}
     Shi \cite{shi}  proved the beautiful result that the number of regions of the type $A_{n-1}$ Shi arrangement is $(n+1)^{n-1}$.  This statement is clearly  deserving of a combinatorial proof; two different bijections proving this result were provided by 
   Stanley and Pak \cite{sta0, sta} and  Athanasiadis and Linusson \cite{ath-lin}.  Our type $C_n$ results can be considered a generalization of the  Athanasiadis-Linusson bijection. In their  work on parking spaces \cite{arm-rei-rho}, Armstrong, Reiner and Rhoades provide another  generalization of the Athanasiadis-Linusson bijection. %We also study statistics naturally arising from our bijection
     
   
     
 We now review the definitions necessary to state our results.
 
 \old{ A sequence $\a=(a_1, a_2, \ldots, a_n) \in [n]^n$ is a \textbf{parking function} if and only if 
 the increasing rearrangement $b_1\leq b_2\leq \cdots \leq b_n$ of $a_1, a_2, \ldots, a_n$ satisfies $b_i \leq i$. Denote by $PF(n)$ the set of all parking functions of length $n$. 
  Let $$\A(n)=\{\a=(a_1, a_2, \ldots, a_n) | a_i \in [n+1], i \in [n]\}.$$ Denote by $d(\a)$ the number of distinct numbers contained in the sequence $\a$.}

   
   \old{Our first description of the type $A_{n-1}$ bijection proving that $|R(\sa)|=(n+1)^{n-1}$ yields a natural correspondence between the multiset $\mathcal{M}$ in which each element of $R(\sa)$ appears $n+1$ times and the sequences $\A(n)$, while the second description, analogous to that of Athanasiadis and Linusson \cite{ath-lin}  is a direct correspondence between $R(\sa)$ and parking functions. The properties of these bijections yield Theorem \ref{gf}.
 
\begin{theorem} \cite{ath-lin} There is a bijection between the regions of $\sa$ and parking functions of length $n$.

\end{theorem}

 \begin{theorem} \label{gf} 
$$ \sum_{R \in R(\sa)} q^{c(R)}= \sum_{R \in R(\sa)} q^{f(R)}=\frac{1}{n+1}\sum_{\textrm{a} \in \A(n)} q^{n-d(\a)}=\sum_{\a \in PF(n)} q^{n-d(\a)}.$$
  \end{theorem}
 
 We use the techniques developed  for the type $A_{n-1}$ case to construct bijective proofs for type $C_n$. 
 }
 
 
 The  \textbf{$C_n$  Coxeter arrangement $\cb$} in $\R^n$ is defined as follows. 
\begin{align*} \cb=\{x_i-x_j=0, x_i+x_j=0, 2x_k=0 \mid 1\leq i<j\leq n, k \in [n]\}.\end{align*}
    
    
    
   The regions of the arrangements $\cb$ naturally correspond to type $C_n$ permutations $w \in \S_n^C$.   Recall that $\S_n^C$ is the group  of all bijections $w$ of the set $[\pm n]=\{-n, -n+1, \ldots, -1, 1, \ldots, n-1, n\}$ onto itself such that $$w(-i)=-w(i),$$ for all $i \in [\pm n]$ and composition as group operation. The notation $w=[a_1, \ldots, a_n]$ means $w(i)=a_i$, for $i \in [n]$, and is called the \textbf{window} of $w$. In \textbf{line notation} this same $w=-a_n\mbox{ } -a_{n-1}\ldots-a_1 \mbox{ } a_1\ldots a_{n-1} \mbox{ }a_n$.
    
    
    
     Let $C^{C}\subset \R^n$ be our  distinguished \textbf{cone} of  $\cc$  corresponding to the type $C_n$ identity permutation: $$C^{C}=\{\x \in \R^n \mid -x_n>-x_{n-1}>\cdots>-x_2>-x_1>x_1>x_2>\cdots>x_{n-1}>x_n\}.$$ 
    
    Let 
    $$ wC^{C}=\{ \x \in \R^n | x_{w(-n)}>x_{w(-n+1)}>\cdots >x_{w(-1)}> x_{w(1)}>x_{w(2)}>\cdots >x_{w(n)} \},$$
    where $\{x_1, \ldots, x_n\}$ are the standard coordinate functions on $\R^n$ and $x_{-i}=-x_i$ for $i <0$. It follows that the number of regions of $\cb$ is $|\mathfrak{S}^C_n|=2^n n!.$ 

    \medskip
    
The      \textbf{type $C_n$  Shi arrangement  $\scc$}  \cite{shi}  is: 
     $$\scc=\cc \cup \{x_i-x_j=1, x_i+x_j=1, 2x_k=1 \mid 1\leq i<j\leq n, k \in [n]\}.$$
     
     
We construct a bijection between the  regions of the  type $C_{n}$  Shi arrangement $\scc$ in $\R^n$ and sequences in the set $$\A^{C}(n)=\{ (a_1, a_2, \ldots, a_n)| a_i \in \{-n, -n+1, \ldots, -1, 0, 1, \ldots, n-1, n\},  i \in [n]\}.$$  

\begin{theorem} \label{thm:bijc} The map $\phi$, which is defined in Section \ref{secc},  is a bijection between the regions of $\scc$ and  sequences in the set $\A^{C}(n)$.
\end{theorem}

Athanasiadis and Linusson \cite[Section 4, Question 3]{ath-lin} were the first to ask for the construction of such  bijection in their paper dealing with the type $A_{n-1}$ case. The properties of our bijection yield Theorem \ref{bgf}. To state it we need a little more terminology.  A hyperplane $H$ is a \textbf{wall} of a region $R$ if it is the affine span of a codimension-1 face of $R$. A wall $H$ is called a \textbf{floor} if $H$ does not contain the origin and $R$ and the origin lie in opposite half-spaces defined by $H$. Denote by $f(R)$ the number of floors of $R$. A wall $H$ is called a  \textbf{ceiling} if $H$ does not contain the origin and $R$ and the origin lie in the same half-spaces defined by $H$. Denote by $c(R)$ the number of ceilings of $R$. Denote by $R(\mathcal{H})$ the set of regions of the hyperplane arrangement $\mathcal{H}$.

 \begin{theorem} \label{bgf} 
$$ \sum_{R \in R(\scc)} q^{c(R)}= \sum_{R \in R(\scc)} q^{f(R)}=\sum_{\a \in \A^{C}(n)} q^{n-d^{C}(\a)},$$
where  $d^{C}(\a)$ is the number of distinct absolute values of the nonzero numbers appearing in ~$\a$.
 \end{theorem}


The outline of the paper is as follows. In Section \ref{sec:poset} we explain the connection between the regions of the type $C_n$  Shi arrangements and the poset of  nonnesting $C_n$-partitions. In  Section \ref{secc} we build on this connection to   prove Theorems \ref{thm:bijc} and \ref{bgf}.   Section  \ref{sec:bb} reiterates the basic thoughts of the paper on the level of posets and sequences. 

        \section{Posets and the regions of $\scc$}
    \label{sec:poset}
    
  
   In this section we explain how to label a region $R$ of  $\scc$ by  the set of its ceilings and the permutation $w \in \S_n^C$, if $R$ is in the cone $wC^{C}$. We will see that  the set of ceilings can be encoded as certain antichains of a special poset. Such an approach is inspired by a correspondence developed by Stanley in \cite[Section 5]{sta0}  between the antichains of a  family of posets   and regions of the type $A_{n-1}$ Shi arrangement.  
   Our bijection between the regions of $\scc$ and sequences (as presented in Section \ref{secc})  will grow out  from  an extension of Stanley's correspondence to the type $C_n$ Shi arrangement.  
%    For a related  bijection  between the positive chambers of the Shi arrangement and order ideals of the root poset of  corresponding type  see \cite[Theorem 5.1.13]{arm} and \cite{cel-pap}.
    For basic definitions about posets  see \cite[Chapter 3]{ec1}.
    
 \old{  \subsection{Posets and the regions of $\sa$.}   \label{sub:aa}  Each region of $\sa$ lies in one of the cones  of ${\rm Cox}^A(n)$. We restrict our attention to the regions of $\sa$ in an arbitrary cone $wC^A$, $w \in \S_n$. Each such region is uniquely determined by the set of its ceilings (or the set of its floors). The set of hyperplanes of $\sa$ intersecting $wC^A$ is 
    $$\H_w=\{x_{w(i)}-x_{w(j)}=1 | 1\leq i<j\leq n, w(i)<w(j)\}.$$
    
    There are two natural orders on the hyperplanes in $\H_w$; namely, hyperplane $H_1$ is less than hyperplane $H_2$ if all the points in $wC^A$ which are on the same  (opposite) side of $H_1$ as the origin are also on the same (opposite) side of $H_2$ as the origin. Thus, $\H_w$ can be considered as a poset. The  set of ceilings of some region of $\sa$ in $wC^A$ is an antichain of this poset. As Theorem \ref{bija} states below, the reverse is also true, and so the antichains of  $\H_w$  are in bijection with the regions of $\sa$ in $wC^A$. To avoid any confusion we now (re-)define the poset we consider.
    
    Let $$Q_w=\{(i, j) : 1\leq i<j\leq n, w(i)<w(j)\}$$  partially ordered by $$(i, j)\leq (r, s) \text{ if } r\leq i<j\leq s.$$   We think of $(i,j) \in Q_w$ as the hyperplane $x_{w(i)}-x_{w(j)}=1$ in $\H_w$.  Note that in $Q_w$  we have  $(i, j)\leq (r, s)$ if and only if 
    all  points in $wC^A$ which are on the same  side of $x_{w(i)}-x_{w(j)}=1$ as the origin are also on the same side of $x_{w(r)}-x_{w(s)}=1$ as the origin.
    
    We represent antichains of $Q_w$ as partitions of $[n]$, where we draw an arc $(i, j)$ in the diagram  if $(i, j) \in Q_w$ is in the antichain. For basic definitions about partitions  see \cite[Chapter 1]{ec1}. Bijecting the regions of $\sa$ to their set of ceilings, and the set of ceilings to the corresponding antichains in $Q_w$, which we represent as partitions,  we obtain a labeling of the regions of $\sa$ by partitions, as shown on Figure \ref{wnumomega}. 
     
     
     
\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.5]{v20i2p31/wnumomega.pdf} 
\caption{Labeling the regions of $\mathcal{S}_3^A$ by partitions.} 
\label{wnumomega} 
\end{center} 
\end{figure}
     
\begin{theorem} \label{bija}
\cite[Section 5]{sta0}, \cite[Theorem 2.1]{sta} 
  There is a bijection between the regions of $\sa$ contained in the cone $wC^A$ and the antichains of the poset $Q_w$. In particular,  \begin{equation} \label{a} |R(\sa)|=\sum_{w \in \S_n} j_P(Q_w), \nonumber \end{equation}
where    $j_P(Q_w)$ denotes the number of antichains of the poset $Q_w$.   \end{theorem} 
     
     \proof It is clear from the above that there is an injective map from  the regions of  $\sa$ to the multiset of the antichains of the posets $Q_w$, 
    $w \in \S_n$.  Since it is known that $|R(\sa)|=(n+1)^{n-1}$ \cite{shi} and  $\sum_{w \in \S_n} j_P(Q_w)=(n+1)^{n-1}$ can be proved without reference to $\sa$ (see Corollary \ref{cor:aa} in Section \ref{sec:aa}) the map also has to be surjective and  Theorem \ref{bija} follows.
     \qed
     
     \medskip
     
     
       Studying the relations of the poset $Q_w$ we see that the antichains of $Q_w$ correspond to nonnesting  $A_{n-1}$-partitions  if we think of $(k, l) \in Q_w$ as an arc in a partition of $[n]$. 

   }
  
  
 Pick a region $R$ of $\scc$ in the cone $wC^{C}$ of $\cb$, $w \in \S_n^C$. The set of hyperplanes of $\scc$ that  intersect $wC^{C}$ is
   $$\H_w^C=\H_w^+\cup \H_w^-$$ %\cup \H_w^c,$$
   where 
  $$\H_w^-= \{ x_{w(i)}-x_{w(j)}=1 \mid i<j, 0<w(i)<w(j)\}$$ 
and  $$\H_w^+= \{x_{w(i)}-x_{w(j)}=1 \mid i<j, w(j)<0<w(i)\}.$$
  
 % and $$\H_w^c= \{x_{w(i)}-x_{w(-i)}=1 \mid i<-i, w(-i)<0<w(i)\}.$$
   
 
 Taking into consideration that $x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}$ since $w(i)=-w(-i)$ and $x_{-i}=-x_i$ for all $i \in [\pm n]$, it follows that    
     $$\H_w^C= \{ x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}=1 \mid i<j, 0<w(i)\leq|w(j)|\}.$$ 
  

   
  \noindent \textbf{Partial order on the hyperplanes.} If $x_{w(a)}-x_{w(b)}=1$, $a<b$, and $x_{w(a')}-x_{w(b')}=1$, $a'<b'$, belongs to $\H_w^C$  and $R$ is on the same side of the hyperplane $x_{w(a)}-x_{w(b)}=1$  as the origin and $a'\leq a<b\leq b',$ then $R$ is also on the same side of the hyperplane $x_{w(a')}-x_{w(b')}=1$ as the origin, since $x_{w(a')}-x_{w(b')}\leq x_{w(a)}-x_{w(b)}<1$. Considering all such implications among the hyperplanes of $\H_w$ we arrive to a partial order (there are two choices of partial order, pick one) on the hyperplanes. Note that if  $x_{w(a)}-x_{w(b)}=1$ is a ceiling of $R$,  then  $x_{w(a')}-x_{w(b')}=1$ cannot be its wall, so cannot be its ceiling either. We will make the convention that the hyperplane  $x_{w(a)}-x_{w(b)}=1$  is {\bf bigger} than the hyperplane $x_{w(a')}-x_{w(b')}=1$ in some poset of hyperplanes, which we formalize below. By the above observations two ceilings are always incomparable in this order. 
  
 \noindent \textbf{A poset based on the partial order on the hyperplanes.}  The following poset could be defined on the set of hyperplanes directly, but for ease of representation we do it otherwise. Define the poset  $Q^C_w$ containing both $(i, j)$ and $(-j, -i)$ subject to constraints below: 
  \begin{equation} \label{poset} Q^C_w=\{(i, j), (-j, -i) \mid  i, j \in [\pm n],  i<j, 0<w(i)\leq|w(j)| \}\end{equation} with the partial ordering inherited from the hyperplanes: \begin{equation} \label{order}(i, j)\leq (r, s) \text{ if } r\leq i<j\leq s.\end{equation} See Figure \ref{fig:poset} for an example.
 
 
\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.7]{v20i2p31/poset.pdf} 
\caption{Poset $Q^C_w$ for $w=[-2, -1]$.} 
\label{fig:poset} 
\end{center} 
\end{figure}
 
 
 
 
  Note that when $(i, j) \in Q^C_w$, then $(-j, -i) \in Q^C_w$, and these two elements are incomparable unless $i=-j$.  In our informal thinking,  these two elements stand for the same hyperplane: $ x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}=1$. 
The reason we include both of these elements in $Q^C_w$, as opposed to just one of them,  is that we do not want to miss any implication among the hyperplanes as explained in the above paragraph``Partial order on the hyperplanes." For example, if $w(1)=1, w(4)=5, w(2)=-3, w(3)=-2$, then both $(1, 4)$ and $(2, 3)$ are in  $Q^C_w$, $(2,3)<(1,4)$, while if we only included one of $(i, j)$ and  $(-j, -i)$ then we could have missed this relation.
   
   The antichains of $Q^C_w$ with the property that if the element $(i, j)$ is in the antichain, then so is $(-j, -i)$, $i, j \in [\pm n]$, correspond to nonnesting $C_n$-partitions if we think of $(k, l) \in Q^C_w$ as an arc in a partition of $[\pm n]$. In the rest of the paper we call  the property that if the element $(i, j)$ is in an antichain, then so is $(-j, -i)$, $i, j \in [\pm n]$, \textbf{property P}.  Recall that a \textbf{nonnesting  $C_n$-partition}  of $[\pm n]$ can be thought of as a nonnesting diagram of arcs, which are drawn over the ground set $-n, -n+1, \ldots, -2, -1, 1, 2, \ldots, n-1, n$ (in this order) such that if there is an arc between $i$ and $j$, for $i, j \in [\pm n]$, then there is also an arc between $-j$ and $-i$ (there are no multiple arcs). See Figure \ref{C-nonnest} for an example.

  The antichains of $Q^C_w$ with the property P are of interest to us, since they encode the sets of the ceilings of the regions.  We can think of mapping a region of $\S_n^C$ to the set of its ceilings  (or  its floors) (more precisely, when talking of a ceiling $ x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}=1$, we are talking of the elements $(i, j)$ and $(-j, -i)$ in $Q_w^C$) to obtain antichains with property P, see Figure \ref{fig:shi_C} and its caption. Whether we consider the set of ceilings or floors, Theorem \ref{bijc} follows. For a related  bijection  between the positive chambers of the Shi arrangement and order ideals of the root poset of  corresponding type  see \cite[Theorem 5.1.13]{arm} and \cite{cel-pap}.
   
  \begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.7]{v20i2p31/shi_C.pdf} 
\caption{We label each region of the type $C_n$ Shi arrangement $\mathcal{S}_n^C$  by a nonnesting  $C_n$-partition and  a type $C_n$ permutation $w$. The permutation $w$ is specified by the cone. Consider the set of ceilings of the regions of $\mathcal{S}_n^C$. Draw the arcs $(i, j)$ and $(-j, -i)$ for the ceilings  $ x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}=1$ obtaining a nonnesting  $C_n$-partition for each region. We drew in these partitions for the cone defined by $x_1>x_2>-x_{2}>-x_{1}$, corresponding to the permutation $w=1 \mbox{ } 2 \mbox{ } -2 \mbox{ } -1$. Note that these partitions exactly correspond to the antichains of $Q_{[ -2, -1]}^C$ (see Figure \ref{fig:poset}) possessing property P.  } 
\label{fig:shi_C} 
\end{center} 
\end{figure}

   
   
      \begin{theorem} \label{bijc}    The regions of $\scc$ contained in $wC^{C}$ are in bijection with the antichains of ~$Q^C_w$ possessing  property P.
   In particular,  \begin{equation} \label{a} |R(\scc)|=\sum_{w \in \S^C_n} j_P(Q^C_w), \nonumber \end{equation}
where    $j_P(Q^C_w)$ denotes the number of antichains of the poset $Q^C_w$ possessing property $P$.   \end{theorem} 
     
     \proof It is clear from the above that there is an injective map from  the regions of  $\scc$ to the multiset of the antichains of the posets $Q^C_w$ possessing property $P$, 
    $w \in \S^C_n$.  Since it is known that $| R(\scc)|={(2n+1)}^{n}$ \cite{shi2} and  $\sum_{w \in \S^C_n} j_P(Q^C_w)=(2n+1)^{n}$ can be proved without reference to $\scc$ (see Corollary \ref{cor:bb} in Section \ref{sec:bb})   the map also has to be surjective and  Theorem \ref{bijc} follows.
     \qed
     
\medskip

%Note that the antichains of $Q^C_w$ correspond to nonnesting $C_n$-partitions   if we think of $(k, l) \in Q^C_w$ as an arc in a partition of $[\pm n]$. Recall that a \textbf{nonnesting  $C_n$-partition}  of $[\pm n]$ can be thought of as a nonnesting diagram of arcs, which are drawn over the ground set $-n, -n+1, \ldots, -2, -1, 1, 2, \ldots, n-1, n$ (in this order) such that if there is an arc between $i$ and $j$, for $i, j \in [\pm n]$, then there is also an arc between $-j$ and $-i$ (there are no multiple arcs). See Figure \ref{C-part} for an example.
   
  
%\begin{figure}[htb] 
%\begin{center} 
%\includegraphics[scale=.85]{v20i2p31/C-part.pdf} 
%\caption{A $C_5$-partition.} 
%\label{C-part} 
%\end{center} 
%\end{figure}
 
\textbf{Labeling the regions of $\mathcal{S}_n^C$.}  Reiterating from above, we label each region of the type $C_n$ Shi arrangement $\mathcal{S}_n^C$  by a nonnesting  $C_n$-partition and  a type $C_n$ permutation $w$. The permutation $w$ is specified by the cone $wC^C$ in which the region lies. The nonnesting  $C_n$-partition is obtained in the following way. Consider the set of all ceilings of a region of $\mathcal{S}_n^C$. Draw the arcs $(i, j)$ and $(-j, -i)$ for the ceilings  $ x_{w(i)}-x_{w(j)}=x_{w(-j)}-x_{w(-i)}=1$ obtaining a nonnesting  $C_n$-partition for each region. These partitions exactly correspond to the antichains of $Q_w^C$ possessing property P. See Figure \ref{fig:shi_C}.

\old{bijecting the regions of $\scc$ to their set of ceilings, and the set of ceilings to the corresponding antichains in $Q^C_w$ possessing property $P$, which we represent as $C_n$-partitions,  we obtain a labeling of the regions of $\scc$ by pairs consisting of a nonnesting $C_n$-partition and a type $C_n$ permutation $w$. See Figure \ref{fig:shi_C}. %, analogously to the type $A_{n-1}$ case.
}

   
     
       

 \old{
  
  \section{Sequences and Shi arrangements in type $A_{n-1}$}
    \label{secaa}
 
 
 In this section we construct a bijection between the regions of $\sa$ and $(n+1)$-tuples of sequences $a_1\ldots a_n$, $a_i \in [n+1]$, for $i \in [n]$, such that every such  sequence appears in exactly one of the  $(n+1)$-tuples.  Furthermore, exactly one among the $n+1$  sequences assigned to a region is a parking function, thereby also leading to a bijection between the regions of $\sa$ and parking functions. The same  bijection  previously appeared in the paper by Athanasiadis and Linusson \cite{ath-lin}. Our exposition makes  the enumeration of the ceiling and floor statistic on the regions on $\sa$ transparent, and that it readily generalizes to bijections in the type $C_n$ case. The ceiling and floor statistics on Shi arrangements was also used and studied by Armstrong and Rhoades in their beautiful paper on the Shi and Ish arrangements \cite{arm-rho}.  The  ideas of this section appear explicitly or implicitly in \cite{ath-lin} and  \cite{arm-rho}.
 
 
  For ease of expfosition we consider $n+1$ copies of the arrangement $\sa$, denoted by $${(\sa)}^{\sqcup (n+1)}:={(\sa)}^{(1)}\sqcup \cdots \sqcup {(\sa)}^{(n+1)},$$ and biject the regions of  ${(\sa)}^{\sqcup (n+1)}$ defined as the  regions of  ${(\sa)}^{(1)}, \ldots, {(\sa)}^{(n)}$ and ${(\sa)}^{(n+1)}$ with the sequences $a_1\ldots a_n$, $a_i \in [n+1]$, for $i \in [n]$. 

 
 The \textbf{type} of an $A_{n-1}$-partition $\pi$ is the integer partition $\l$ whose parts are the sizes 
  of the blocks of $\pi$. 

\begin{theorem} \label{a1} \cite{ath} There is a bijection $b$ between the set of type $\l=(\l_1, \ldots, \l_d)$ nonnesting $A_{n-1}$-partitions and pairs $(S, g)$, where $S$ is a $d$-subset of $[n]$ and the map $g:S\rightarrow \{\l_1, \ldots, \l_d\}$ is such that $|g^{-1}(i)|=r_i$, $0 \leq i$. 

\end{theorem}

\proof  Given a  type $\l=(\l_1, \ldots, \l_d)$ nonnesting $A_{n-1}$-partition, let $S$ be the leftmost elements of its blocks. Let $g(s)=k$ if  $s \in S$ is in a block of size $k$. It can be shown by induction on $n$ that  the set $S$ and function $g$ defined this way uniquely determine the nonnesting partition they came from. \qed



\medskip

Label each region of $\san$ by the nonnesting $A_{n-1}$-partition corresponding to an antichain of $Q_w$, $w \in \S_n$, as described in Section \ref{sub:aa} and shown on Figure \ref{"n+1"}. Each region of  $\san$ is completely specified by a number $k \in [n+1]$ (specifying which copy of $\sa$  we are in in $\san$),  a permutation $w \in \S_n$ (specifying the cone of $\sa$), and a nonnesting $A_{n-1}$-partition $\pi$ (specifying the ceilings of $R$ in $wC^A$). While we generally think of $\pi$ as on the vertices $1, 2, \ldots, n-1, n$, in this order, the $A_{n-1}$-partition $\pi$ also has \textbf{$w$-labels} $ w(1), \ldots, w(n-1), w(n)$. See Figure \ref{wlabel}. 



\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.2]{v20i2p31/n+1.pdf} 
\includegraphics[scale=.2]{v20i2p31/n+1.pdf} 
\includegraphics[scale=.2]{v20i2p31/n+1.pdf} 
\includegraphics[scale=.2]{v20i2p31/n+1.pdf} 
\caption{Labeling of the regions of $(\mathcal{S}_3^A)^{\sqcup 4}$ by nonnesting partitions.} 
\label{"n+1"} 
\end{center} 
\end{figure}

\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.5]{v20i2p31/wlabel.pdf} 
\caption{The $w$-labels of partitions labeling of the regions of $\mathcal{S}_3^A$ are shown for the permutations $132$ and $321$.} 
\label{wlabel} 
\end{center} 
\end{figure}   

\begin{lemma} \label{a2}The number of regions of $\sa$ containing the nonnesting $A_{n-1}$-partition $\pi$ of type $\l$ is equal to 
\begin{equation} \label{acc} {n \choose {\l_1, \ldots, \l_d}}. \end{equation}
\end{lemma}

\proof In this proof we effectively count the number of  permutations $w\in \S_n$ such that $\pi$ is an antichain in the poset $Q_w$, since the latter is equal to the of number regions of $\sa$ containing the nonnesting $A_{n-1}$-partition $\pi$. Given a nonnesting $A_{n-1}$-partition $\pi$ of type $\l$ there are   ${n \choose {\l_1, \ldots, \l_d}}$ ways to choose  the values of the $w$-labels which go into the blocks of $\pi$. Since in each  block the $w$-labels  increase,  equation \eqref{acc} follows. 
\qed


\medskip


Given a type $\l=(\l_1, \ldots, \l_d)$ nonnesting $A_{n-1}$-partition  $\pi$, denote by $S_{\pi}$ the set and $g_{\pi}$ the function   from Theorem \ref{a1}. Let $M_{\pi}$ be the multiset consisting of $\l_i$  copies of each element of $g_{\pi}^{-1}(\l_i)$, for each part in the set (not multiset!) $\{\l_1, \ldots, \l_d\}$. An \textbf{$n$-shifted permutation} of $M_{\pi}$ is a permutation of the elements of the multiset $M_{\pi}$ such that each  entry is increased by $k \in \{0, 1, \ldots, n\}$ and taken modulo $n+1$. For example the $2$-shifted permutations of $\{\{1, 2\}\}$ are $12, 21, 23, 32, 31, 13$.

\begin{theorem} \label{a3} There is a bijection $\phi$ between the regions of $\san$ labeled by the nonnesting $A_{n-1}$-partition $\pi$ of type $\l$ and $n$-shifted permutations of the multiset $M_{\pi}$. 

\end{theorem}

\proof There are multiple ways to set up this bijection. We present one way here and note how to define a family of bijections satisfying Theorem \ref{a3}. 

Given  the nonnesting $A_{n-1}$-partition $\pi$, a  permutation $w$ for which $\pi$ is an antichain in $Q_w$, and an integer  $k \in [n+1]$ specifying which copy of $\sa$  we are in in $\san$, order the blocks  of $\pi$  by increasing size. The blocks of the same size are ordered lexicographically according to the $w$-labels on them. Order the numbers in the multiset  $M_{\pi}$ so that the numbers with less multiplicites are smaller. Among the numbers with the same multiplicity order them according to the natural order on integers. The previous two orders yield a correspondence $f$ between the blocks $B$ of $\pi$ and the numbers from $M_{\pi}$.  (This correspondence could of course  be defined in several ways leading to different bijections.) Let the values of the $w$-labels of $B$ specify the positions that the number $f(B)+k-1 \textrm{ mod } n+1$ is taking.  


Correspondence $f$ could also be naturally defined by the bijection given in Theorem \ref{a1}. As it turns out both descriptions of $f$  in the type $A_{n-1}$ case are the same.


Figure \ref{fig:bij} shows the construction of the bijection on $(\mathcal{S}^A_3)^{(1)}$. For example, consider the region $R$ in $(132)C^A$ with ceiling $x_1-x_3=x_{w(1)}-x_{w(2)}=1$.  The $w$-labels are written above the partition and the numbers corresponding to the elements of the blocks are below the partition and are circled individually. Then, to get the sequence corresponding to the region, read the circled numbers in the order specified by the $w$-labels. The resulting $3$-digit sequence is circled on  Figure \ref{fig:bij}.  



\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.5]{v20i2p31/bij.pdf} 
\caption{Constructing the bijection on $(\mathcal{S}^A_3)^{(1)}$.} 
\label{fig:bij} 
\end{center} 
\end{figure}


For the restriction of the bijection to $(\mathcal{S}^A_3)^{(1)}$ see Figure \ref{bijcomplete}.


\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.5]{v20i2p31/bijcomplete.pdf} 
\caption{The bijection on $(\mathcal{S}^A_3)^{(1)}$.} 
\label{bijcomplete} 
\end{center} 
\end{figure}



The above defined map is a  bijection  between the regions of $\san$ labeled by the nonnesting $A_{n-1}$-partition $\pi$ of type $\l$ and  $n$-shifted permutations of the multiset $M_{\pi}$, which can be shown by writing down an explicit inverse, or by noting that it is injective and the domain and codomain are equinumerous.
\qed

\medskip


Extend the map $\phi$ defined in the proof of  Theorem \ref{a3} to a map between 
all regions of $\san$ and the set of sequences $\A(n)=\{a_1\ldots a_n | a_i \in [n+1], i \in [n]\}$.



\begin{theorem} \label{a4} (cf. \cite{ath-lin})  The map  $\phi: R(\san)\rightarrow \A(n)$ is a bijection.  
\end{theorem} 


\begin{theorem} \label{a5} (cf. \cite{ath-lin})  The restriction of the bijection $\phi$ to the first copy of the Shi arrangement ${\sa}^{(1)}$ is a bijection between the regions of the Shi arrangement and parking functions.\end{theorem} 

We leave the details of the proofs of Theorems \ref{a4} and \ref{a5} to the reader. Hint: see \cite[Exercise 5.49]{ec2}. 

\begin{corollary} 
\begin{equation} \label{uhhu}
  \sum_{\l \vdash n} {n+1 \choose d} \frac{d!}{m_{\l}}{n \choose {\l_1, \l_2, \ldots, \l_d}}=(n+1)^n,\end{equation} 
where $m_{\l}=\prod_{i=1}^n r_i!$, if $r_i$ denotes the number of parts of $\l$ equal to $i$.
   \end{corollary}


\proof Kreweras 
\cite[Theorem 4]{kre}  proved that the number of noncrossing partitions of $[n]$ of type $\l$ is equal  $$ \frac{n!}{m_{\l}(n-d+1)!},$$ 
  where $d$ denotes the number of parts of $\l$. Athanasiadis \cite[Theorem 3.1]{ath} gave a bijection between noncrossing and nonnesting $A_{n-1}$-partitions which preserves type. Thus, the total number of nonnesting $A_{n-1}$-partitions of type $\l$ labeling the regions of $\san$ is 
 \begin{equation}\label{n+1} (n+1)\frac{n!}{m_{\l}(n-d+1)!}= {n+1 \choose d} \frac{d!}{m_{\l}}.\end{equation}
 Equation \eqref{n+1} together with Lemma \ref{a2} and Theorem \ref{a4} imply equation \eqref{uhhu}. 
\qed

\medskip

Theorem \ref{gf} from the introduction is  a corollary of the proofs of Theorems \ref{a3}, \ref{a4} and \ref{a5}.  For further details see Section \ref{sec:aa}, and in particular Theorem \ref{bij}.
\medskip

\noindent \textbf{Theorem 2.}
$$ \sum_{R \in R(\sa)} q^{c(R)}= \sum_{R \in R(\sa)} q^{f(R)}=\frac{1}{n+1}\sum_{\textrm{a} \in \A(n)} q^{n-d(\a)}=\sum_{\a \in PF(n)} q^{n-d(\a)}.$$
 
 
}

    \section{Sequences and Shi arrangements in type $C_{n}$}
    \label{secc}


 In this section we construct a  bijection  between the regions of  $\scc$ and the set of  sequences $\A^{C}(n)=\{a_1\ldots a_n|  a_i \in  [\pm n] \cup \{0\}, i \in [n]\}$.   Our proof  yields  enumeration of regions by  the ceiling and floor statistic, which we express in a generating function form. 
  
 The \textbf{type} of a $C_{n}$-partition $\pi$ is the integer partition $\l$ whose parts are the sizes 
  of the nonzero blocks of $\pi$, including one part for each pair of blocks $\{B, -B\}$.   The  zero block is a block $B$ such that $B=-B$. Figure \ref{C-nonnest} shows a nonnesting $C_5$-partition with blocks $\{2\}, \{-2\}, \{-1, -4\}, \{1, 4\}, \{-5,-3,$ $ 3, 5\}$.  The last block is a zero block, and so the type of this partition is $(2, 1)$. 
  
  
  
\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=.85]{v20i2p31/C-nonnest.pdf} 
\caption{A type $(2, 1)$ nonnesting $C_5$-partition.} 
\label{C-nonnest} 
\end{center} 
\end{figure}
  
  
Recall from Section \ref{sec:poset} that we label each region of $\scc$ by the nonnesting $C_n$-partition corresponding to an antichain of $Q_w^C$, $w \in \S_n^C$, possessing property $P$,  and a permutation $w \in \S_n^C$. While we generally think of $\pi$ as on the vertices $-n, -n+1, \ldots, -1, 1, 2, \ldots, n-1, n$, in this order, the $C_n$-partition $\pi$ also has \textbf{$w$-labels} $w(-n), w(-n+1), \ldots, w(-1)$, $w(1),\ldots, w(n-1), w(n)$. Given a block $B=\{v_1,\ldots, v_k\}$ of $\pi$  the set of $w$-labels of $B$ is $\{w(v_1),\ldots, w(v_k)\}$.



\begin{lemma} \label{w}
Given a nonnesting $C_n$-partition $\pi$, let    $S_B$ be a set of size $|B|$ for each block $B$ of $\pi$ such that the sets $S_B$'s are disjoint, their union is $[\pm n]$ and $S_B=-S_{-B}$, where $-S_{-B}=\{-a \mid a \in S_{-B} \}$. Then there exists a unique  $w$ such that $\pi$ is an antichain in $Q_w^C$ possessing property $P$ and the set of $w$-labels of each block $B$ are equal to $S_B$.\end{lemma}

\proof Recall that $$Q^C_w=\{(i, j), (-j, -i) \mid   i,j \in [\pm n], i<j, 0<w(i)\leq|w(j)| \}$$ with the partial ordering: $$(i, j)\leq (r, s) \text{ if } r\leq i<j\leq s.$$

Suppose that nonnesting $C_n$-partition $\pi$ is an antichain in $Q_w^C$ possessing property $P$.  Then, for a block $B=\{v_1, v_2, \ldots, v_k\}$ of   $\pi$  with $v_1<v_2<\cdots<v_k$,  it must be that for all $l \in [k-1]$ either \begin{equation} \label{peak} (0<w(v_l)\leq |w(v_{l+1})|)\mbox{ or }(w(v_{l+1})<0\mbox{ and  } |w(v_{l+1})|\leq |w(v_l)|).\end{equation} This follows since by the definition of $Q_w^C$ either $(v_l, v_{l+1})$ satisfies  $0<w(v_l)\leq|w(v_{l+1})|$ or $(-v_{l+1},-v_l)$ satisfies $0<w(-v_{l+1})\leq|w(-v_{l})|$.   Note that \eqref{peak} implies that the $w$-labels of $B$ are such that all the positive $w$-labels come first followed by all the negative $w$-labels and the absolute values of the $w$-values read from left to right form a unimodal sequence. Since there is a unique way to arrange a set of numbers in this manner, and when arranged so $\pi$ is an antichain in $Q_w^C$ possessing property $P$, it follows that $w$ is unique. See Figure \ref{fig:peak} for an example.
\qed


\begin{figure}[htb] 
\begin{center} 
\includegraphics[scale=1.]{v20i2p31/blocks_determine.pdf} 
\caption{The blocks of the nonnesting $C_n$-partition $\pi$ presented above are $B_0=\{-1, 1\}$, $B_1=\{-2\}$, $B_2=\{-3,-4,-5\}$. For sets $S_{B_0}=\{-5, 5\},$ $S_{B_1}=\{3\}$ and $S_{B_2}=\{-1, 2, 4\},$ as described in Lemma \ref{w},  the unique $w$ for which  $\pi$ is an antichain of $Q_w^C$ is $w=[-5, -3, 1, -4, -2]$ as shown on the figure in line notation (and its construction explained in the proof of Lemma \ref{w}).} 
\label{fig:peak} 
\end{center} 
\end{figure}
     


\begin{lemma} \label{2} In the labeling described in Section \ref{sec:poset} the number of regions of $\scc$ labeled by the nonnesting $C_n$-partition $\pi$ of type $\l$ (and some permutation) is equal to 
\begin{equation} \label{cc} {n \choose {\l_1, \ldots, \l_d, n-|\l|}} \prod_{i=1}^d 2^{\l_i}. \end{equation}
\end{lemma}

\proof In this proof we count the number of signed permutations $w\in \S_n^C$ such that $\pi$ is an antichain in the poset $Q_w^C$ possessing property $P$, since the latter is equal to the number of regions of $\scc$ labeled by the nonnesting $C_n$-partition $\pi$. By Lemma \ref{w}  if we have a collection of sets $S_B$  of size $|B|$ for each block $B$ of $\pi$ such that the sets $S_B$'s are disjoint, their union is $[\pm n]$ and $S_B=-S_{-B}$, where $-S_{-B}=\{-a \mid a \in S_{-B} \}$, then there is a unique $w$ for which  $\pi$ is an antichain in $Q_w^C$ possessing property $P$ and the set of $w$-labels of each block $B$ are equal to $S_B$. Moreover, note that for any $w$ for which $\pi$ is an antichain in the poset $Q_w^C$ possessing property $P$  the sets of $w$-labels of the blocks have to satisfy the above criteria for the $S_B$'s. Thus, number of regions of $\scc$ containing the nonnesting $C_n$-partition $\pi$ of type $\l$ is equal to the number of collections of sets $S_B$'s satisfying the above criteria. There are  $ {n \choose {\l_1, \ldots, \l_d, n-|\l|}}$ ways of choosing the absolute values of the elements of the sets $S_B$'s and there are $\prod_{i=1}^d 2^{\l_i}$ ways of choosing the signs for the sets corresponding to the nonzero blocks of $\pi$. Thus, the total number of collections of sets $S_B$'s satisfying the above criteria is ${n \choose {\l_1, \ldots, \l_d, n-|\l|}} \prod_{i=1}^d 2^{\l_i}.$ \qed


  
 The following theorem is based on  a bijection of Fink and Iriarte \cite{fin-iri}  between noncrossing and nonnesting $C_n$-partitions which preserves type and a bijection of Athanasiadis \cite{ath} between noncrossing $C_n$-partitions and pairs $(S, g)$, where $S$ is a set and $g$ is a function subject to the conditions stated below. We invite the interested reader to learn about these bijections from the original papers themselves, as their description would take up considerable amount of space in the current paper and as such it is omitted.  
 


\begin{theorem} \label{1}There is a bijection between the set of type $\l=(\l_1, \ldots, \l_d)$ nonnesting $C_n$-partitions and pairs $(S, g)$, where $S$ is a $d$-subset of $[n]$ and the map $g:S\rightarrow \{\l_1, \ldots, \l_d\}$ is such that $|g^{-1}(i)|=\#\{j \mid \l_j=i, j \in [d]\}$, $0< i$. 
\end{theorem}

\proof \cite[Theorem 2.4]{fin-iri} establishes a type-preserving bijection $b_1$ between nonnesting and noncrossing $C_n$-partitions, and the proof of  \cite[Theorem 2.3]{ath} provides a bijection $b_2$ between the set of type $\l=(\l_1, \ldots, \l_d)$ noncrossing $C_n$-partitions and pairs $(S, g)$, where $S$ is a $d$-subset of $[n]$ and the map $g:S\rightarrow \{\l_1, \ldots, \l_d\}$ is such that $|g^{-1}(i)|=\#\{j \mid \l_j=i, j \in [d]\}$, $0< i$. \qed






\old{
\begin{lemma} \label{2}The number of regions of $\scc$ containing the nonnesting $C_n$-partition $\pi$ of type $\l$ is equal to 
\begin{equation} \label{cc} {n \choose {\l_1, \ldots, \l_d, n-|\l|}} \prod_{i=1}^d 2^{\l_i}. \end{equation}
\end{lemma}

\proof In this proof we effectively count the number of signed permutations $w\in \S_n^C$ such that $\pi$ is an antichain in the poset $Q_w^C$ possessing property $P$, since the latter is equal to the number of regions of $\scc$ containing the nonnesting $C_n$-partition $\pi$. Given a nonnesting $C_n$-partition $\pi$ of type $\l$ there are   ${n \choose {\l_1, \ldots, \l_d}, n-|\l|}$ ways to choose  the absolute values of the $w$-labels which go into the blocks of $\pi$. Let $2(n-|\l|)$ be the  size of the zero block of $\pi$. The signs and order of the $w$-labels in the zero block of $\pi$ are determined: there have to be $(n-|\l|)$ positive numbers in increasing order followed by their $(n-|\l|)$ negatives in increasing order. Each nonzero  block is comprised  of  a possibly empty sequence of positive $w$-labels in increasing order followed by a possibly empty sequence of negative $w$-labels in increasing order. There are exactly $2^{\l_i}$ ways to decide the signs among $\l_i$ numbers, and once the signs are decided so is the order. Thus, equation \eqref{cc} follows. 
\qed
}



Given a type $\l=(\l_1, \ldots, \l_d)$ nonnesting $C_n$-partition  $\pi$, denote by $S_{\pi}$ the set and $g_{\pi}$ the function   from Theorem \ref{1}. Let $M_{\pi}$ be the multiset consisting of $n-|\l|$ $0$'s, and $\l_i$  copies of each element of $g_{\pi}^{-1}(\l_i)$, for each part in the set (not multiset!) $\{\l_1, \ldots, \l_d\}$. A \textbf{marked permutation} of $M_{\pi}$ is a permutation of the elements of the multiset $M_{\pi}$ such that each  nonzero  entry has a $\pm$ sign in addition. For example the marked permutations of the multiset $\{\{0, 1, 1\}\}$ are $011, 101, 110, 0-1-1, -10-1, -1-10, 01-1, 10-1, 1-10, 0-11, -101, -110$ (we omitted the $+$ signs).


Given two blocks $B_1$ and $B_2$ in a partition, block $B_1$ is smaller than $B_2$ in the \textbf{order $\sigma$} if the smallest vertex that  $B_1$ contains is smaller than the smallest vertex that $B_2$ contains. By convention, if for a block $B\neq -B$, we consider block $B$  smaller than block $-B$ in the order $\sigma$. 

\begin{theorem} \label{3} There is a bijective map $\phi$ between the regions of $\scc$ labeled by the nonnesting $C_n$-partition $\pi$ of type $\l$ and marked permutations of the multiset $M_{\pi}$. 
\end{theorem}

\proof There are multiple ways to set up the map $\phi$. We present one of these ways here, and based on it the interested reader can device several others (though one will of course suffice for all we need it).
  
Given  the nonnesting $C_n$-partition $\pi$ of type  $\l=(\l_1, \ldots, \l_d)$ first we define a map $f$ from the  pairs of blocks ($\{B, -B\}$) of $\pi$ to the set  underlying $M_{\pi}$. If $\pi$ has a zero block $B=-B$, then let $f(B)=0$. 
Let $\{B_1, -B_1, B_2, -B_2, \ldots, B_d, -B_d\}$ be all the nonzero blocks of $\pi$, where $|B_i|=\l_i$,  $i \in [d]$, such that if $\l_j=\l_{j+1}$ for some $j \in [d-1]$, then $B_j$ is smaller than $B_{j+1}$ in the order $\sigma$. 

 Order the nonzero numbers in the multiset  $M_{\pi}$ so that  the numbers with bigger multiplicites come first. Among the numbers with the same multiplicity order them according to the natural order on integers. Note that by construction the sequence we get as a result has the form $a_1^{\l_1}a_2^{\l_2} \ldots a_d^{\l_d}$, where $a_i^{\l_i}$ denotes the sequence of $\l_i$ $a_i$'s.
Let $f(B_i)=a_i$ for $i \in [d]$.


Recall that given a region $R$ of $\scc$ it is  labeled by the nonnesting $C_n$-partition $\pi$ of type $\l$ and a signed permutation $w$ for which $\pi$ is an antichain in $Q_w^C$ possessing property $P$. For such a region $R$  construct the marked permutation $\phi(R)=c_1\ldots c_n$  of the multiset $M_{\pi}$ by the following rule: if vertex $v$ is in block $B_k$, $k \in \{0\} \cup [d]$, then $c_{|w(v)|}=sign(w(v))f(B_k)$, where $sign(a)=-1$ if $a<0$ and $sign(a)=1$ if $a>0$. 

To show that $\phi$ is bijective, we exhibit its inverse. Given a marked permutation $c_1\ldots c_n$  of the multiset $M_{\pi}$, we trivially obtain the underlying multiset and from that we can obtain the set-function pair $(S_{\pi}, g_{\pi})$  from which the multiset was constructed. From these, by Theorem \ref{1} we  can recover  the (unique)  $C_n$-partition $\pi$ associated to the region(s). Now, knowing the blocks of $\pi$ and the multiset $M_{\pi}$, we can reconstruct the function $f$. Once we know $f$, the marked permutation $c_1\ldots c_n$ specifies the set of $w$-labels on each block of $\pi$ and  by Lemma \ref{w} that uniquely specifies $w$. Thus, there is a unique region - namely the one labeled by $\pi$ and $w$ - which maps to $c_1\ldots c_n$ under $\phi$.
\qed
\old{order the (pair of) blocks $\{B, -B\}$ of $\pi$ as follows. If there is a zero block, then it comes first. The other blocks are ordered by increasing size ($|B|$), and the blocks of the same size are ordered lexicographically according to the $w$-labels on them (on $B$s). Order the numbers in the multiset  $M_{\pi}$ so that the $0'$s come first, and among the other numbers the numbers with less multiplicites are smaller. Among the numbers with the same multiplicity order them according to the natural order on integers. The previous two orders yield a correspondence $f$ between the blocks $B$ of $\pi$ and the numbers from $M_{\pi}$.  (This correspondence could of course  be defined in several ways leading to different bijections.) Let the absolute values of the $w$-labels of $B$ specify the positions that the number $f(B)$ is taking. For the $w$-labels of nonzero blocks which are negative add a $-$ to the number in the corresponding spot.

Correspondence $f$ could also be naturally defined by the bijection given in Theorem \ref{1}. 



The above defined maps are  bijections  between the regions of $\scc$ labeled by the nonnesting $C_n$-partition $\pi$ of type $\l$ and marked permutations of the multiset $M_{\pi}$, which can be shown by writing down explicit inverses, or by noting that they are injective and the domains and codomains are equinumerous.}




\medskip


Extend the map $\phi$ defined in the proof of  Theorem \ref{3} to a map between 
all regions of $\scc$ and the set of sequences $\A^{C}(n)=\{a_1\ldots a_n | a_i \in [\pm n]\cup \{0\}, i \in [n]\}$, to obtain the following corollaries.% as in the type $A_{n-1}$ case.



\begin{theorem} \label{4} The map  $\phi: R(\scc)\rightarrow \A^{C}(n)$ is a bijection.  
\end{theorem} 



\begin{corollary} 
$$\sum_{\l \vdash n} \frac{n!}{m_{\l}(n-d)!} {n \choose {\l_1, \ldots, \l_d, n-|\l|}} \prod_{i=1}^d 2^{\l_i}=(2n+1)^n,$$  where $m_{\l}=\prod_{i=1}^n r_i!$, if $r_i$ denotes the number of parts of $\l$ equal to $i$.
\end{corollary}


\proof
Athanasiadis \cite{ath} proved that the number of  nonnesting   $C_n$-partitions  of type $\l$ is  $$ \frac{n!}{m_{\l}(n-d)!},$$ which together with Lemma \ref{2} and Theorem \ref{4} imply the above equality. 
\qed

\medskip

Theorem \ref{bgf} is   a corollary of the proofs of  Theorems \ref{1}, \ref{3} and  \ref{4}. For further details see Section \ref{sec:bb}, and in particular Theorem \ref{cbij}.

\medskip


\noindent \textbf{Theorem 2.}
$$ \sum_{R \in R(\scc)} q^{c(R)}= \sum_{R \in R(\scc)} q^{f(R)}=\sum_{\a \in \A^{C}(n)} q^{n-d^{C}(\a)}.$$
 

\old{ \section{Posets and sequences in  type $A_{n-1}$}
    \label{sec:aa}
    
    
In this section we revisit the type $A_{n-1}$ world of  posets $Q_w$, $w \in \S_n$, and parking functions of length $n$ and state their relation explicitly without the mention of arrangements. Much of the considerations of this section appear in the work of Athanasiadis and Linusson \cite{ath-lin} and Armstrong and Rhoades \cite{arm-rho} either explicitly or implicitly. We highlight our perspective on the relation of the posets and sequences and study their properties in detail.  We carry out a similar agenda for the posets $Q^{C}_w$, $w \in \S_n^C$, and sequences in $\A^{C}(n)$ in the next section.
     
       
 Recall that $$Q_w=\{(i, j) : 1\leq i<j\leq n, w(i)<w(j)\}$$  is partially ordered by $$(i, j)\leq (r, s) \text{ if }r\leq i<j\leq s.$$ We explore the refinements of the equation
 \begin{equation} \label{q} \sum_{w \in \S_n} j_P(Q_w)=(n+1)^{n-1},\end{equation}
 which follows from  Theorems \ref{bija} and \ref{a4}.  In the process we reiterate the  proof of equation \eqref{q} without reference to arrangements. 

 Partition the set of parking functions of length $n$, $PF(n)$, according to the cardnality of the set $\{a_1, a_2, \ldots, a_n\}$. Let $S_k(n)=\{(a_1, a_2, \ldots, a_n) \in PF(n) : |\{a_1, a_2, \ldots, a_n\}|=k\}$. Then
 $$PF(n)=\bigcup_{k=1}^n S_k(n).$$
 
  
 Partition the multiset of antichains $\mathcal{M}(n)$ of $Q_w$, $w \in \S_n$, according to the cardinality of the antichains.   Let $M_k(n)=\{\{ \{(i_1, j_1), \ldots, (i_k, j_k)\} \in \mathcal{M}(n) \}\}$. Then
 $$\mathcal{M}(n)=\bigcup_{k=0}^{n-1} M_k(n).$$

The following theorem can be deduced  from the work of Athanasiadis and Linusson \cite{ath-lin} and Armstrong and Rhoades \cite{arm-rho}. 

\begin{theorem} \label{bij}
 $$|S_k(n)|=|M_{n-k}(n)|, \mbox{ } k \in [n].$$ 
 \end{theorem}
 
We prove Theorem \ref{bij} by providing a bijection between the sets $S_k(n)$ and $M_{n-k}(n)$, $k \in [n]$. Before proceeding to the proof of  Theorem \ref{bij}   we partition the sets  $S_k(n)$ and $M_{n-k}(n)$, $k \in [n]$, further. 
 
 Partition the set of parking functions of length $n$ with $k$ distinct numbers $S_k(n)$, $k \in [n]$, according to the $k$ distinct numbers appearing in the  sequence $(a_1, a_2, \ldots, a_n)$, and the number of times they appear. If $\{a_1, a_2, \ldots, a_n\}=\{c_1<c_2<\cdots<c_k\}$ and $c_i$ appears $o_i$ times in $(a_1, a_2, \ldots, a_n)$, $ i \in [k]$, let $$S_k^{\c, \o}(n)=\{(a_1, a_2, \ldots, a_n) \in S_k(n) |  
 \{\{a_1, a_2, \ldots, a_n\}\}=\cup_{i=1}^{k}\cup_{j=1}^{o_i}\{\{c_i\}\}  \mbox{ } \},$$
 where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i=n$.
 
 Given an antichain $\aa=\{(i_1, j_1), \ldots, (i_{n-k}, j_{n-k})\} \in M_{n-k}(n)$, $k \in [n]$, it naturally corresponds to  a nonnesting partition $\pi_{\aa}$  of $[n]$ with $k$ blocks,  where the arc diagram of $\pi_{\aa}$ consists of the arcs $(i_1, j_1), \ldots, (i_{n-k}, j_{n-k})$. Order the $k$ blocks of $\pi_{\aa}$ according to their smallest elements $c_i$, $i \in [k]$,  $c_1< \ldots< c_k$. Let $o_i>0$, $i \in  [k]$, be the number of elements in the $i^{th}$ block of $\pi_{\aa}$. Denote by $c(\aa)=(c_1< \ldots< c_k)$ and  $o(\aa)=(o_1, \ldots, o_k)$.
 Partition the multiset of antichains of length $n-k$ of  $Q_w$, $w \in \S_n$, $M_{n-k}(n)$, $k \in [n]$, according to $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i=n$, as described above. Let 
$$M_{n-k}^{\c, \o}(n)=\{\{\aa=\{(i_1, j_1), \ldots, (i_{n-k}, j_{n-k})\} \in M_{n-k}(n)|  
c(\aa)=\c, o(\aa)=\o \}\},$$
where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i=n$.


\begin{lemma} \label{un} \cite{ath}
The vectors $c(\aa)=\c$ and $o(\aa)=\o$, where  $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $k \in [n]$, $o_i>0$, for $i\in [k]$, $\sum_{i=1}^k o_i=n$, $c_1=1$, and $c_i \in \{c_{i-1}+1, c_{i-1}+2, \ldots, o_1+\cdots+o_{i-1}+1\}$, for $i\in \{2, \ldots, k\}$,  uniquely determine the antichain $\aa$.
\end{lemma}
 

 The following theorem can be deduced from the work of Athanasiadis and Linusson \cite{ath-lin} and Armstrong and Rhoades \cite{arm-rho}. 
 
 \begin{theorem} \label{co}
 $$|S_k^{\c, \o}(n)|=|M_{n-k}^{\c, \o}(n)|={ n \choose {o_1, \ldots, o_k}}, $$
 where $k \in [n]$, $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i=n$.
 \end{theorem}
  
  \proof A bijective proof  can be given using Theorem \ref{a1} and the ideas of Theorem \ref{a3}. The enumeration is in Lemma \ref{a2}.  Note that arrangements do not enter the proof. \qed
    \medskip
  
 \noindent \textit{Proof of Theorem \ref{bij}.} Straightforward corollary of Theorem \ref{co}, since 
$$S_{k}(n)=\sum_{\c, \o} S_{k}^{\c, \o}(n)=\sum_{\c, \o} M_{n-k}^{\c, \o}(n)=M_{n-k}(n),$$  where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $k \in [n]$, $o_i>0$, for $i\in [k]$, $\sum_{i=1}^k o_i=n$.
 \qed
 
 
 \begin{corollary} \label{cor:aa}
$$ \sum_{w \in \S_n} j_P(Q_w)=(n+1)^{n-1}.$$
 \end{corollary}
 
 \proof Theorems \ref{co} and  \ref{bij} extend to a bijection between $$MA(n)=\cup_{k=1}^n \cup_{\c, \o}M_{n-k}^{\c, \o}(n) \mbox{ and }PF(n)=\cup_{k=1}^n \cup_{\c, \o}S_{k}^{\c, \o}(n),$$ the cardinalities of which are  $ \sum_{w \in \S_n} j_P(Q_w)$ and $(n+1)^{n-1}$, respectively.
 \qed
 }
 
 \section{Posets and sequences in type  $C_{n}$}
    \label{sec:bb}
    
    
In this section we revisit the type   $C_{n}$ world of  posets $Q^{C}_w$, $w \in \S_n^{C}$, and  sequences in $\A^{C}(n)$  and state their relation explicitly. % without the mention of arrangements.       
       
       Recall that      
$$Q^C_w=\{(i, j), (-j, -i) \mid  i, j \in [\pm n],  i<j, 0<w(i)\leq|w(j)| \}$$       
         is partially ordered by $$(i, j)\leq (r, s) \text{ if }r\leq i<j\leq s.$$ We  will prove refinements of the equation
 \begin{equation} \label{qc} \sum_{w \in \S_n^C} j_P(Q^{C}_w)=(2n+1)^{n},\end{equation}
 without reference to arrangements. Here $j_P(Q^{C}_w)$ denotes the number of antichains of $Q^{C}_w$ possessing property $P$.

Partition $\A^{C}(n)$ according to the number of nonzero absolute
values in the set $\{a_1$, $a_2\ldots, a_n\}$, denoted by $d^{C}(\a)$
for $\a=(a_1,a_2,\ldots, a_n)$. Let $A^{C}_k(n)=\{(a_1, a_2, \ldots,
a_n) \in \A^{C}(n) : d^{C}(\a)=k\}$. Then
 $$\A^{C}(n)=\bigcup_{k=0}^n A^{C}_k(n).$$
 
  
Let $\mathcal{M}^{C}(n)$ be the multiset of antichains of $Q_w^C$ possessing property $P$, $w \in \S_n^C$.  Given an antichain $\x \in  \mathcal{M}^{C}(n)$   it naturally corresponds to  a nonnesting $C_n$-partition $\pi_{\x}$ obtained by simply drawing an arc $(a, b)$ for each $(a, b) \in \x$. Partition the multiset $\mathcal{M}^{C}(n)$ according to the number of pairs    of nonzero blocks in the corresponding nonnesting $C_n$-partition.  Denote by $b(\x)$ the number of pairs of nonzero blocks  in $\pi_{\x}$, $\x \in  \mathcal{M}^{C}(n)$.  Let $M^{C}_k(n)=\{\{ \x \in \mathcal{M}^C(n) | b(\x)=k \}\}$. Then
 $$\mathcal{M}^{C}(n)=\bigcup_{k=0}^{n} M^{C}_k(n).$$

\begin{theorem} \label{cbij}
 $$|A^{C}_k(n)|=|M^{C}_{k}(n)|, \mbox{ } k \in \{0\}\cup [n].$$ 
 \end{theorem}
 
We prove Theorem \ref{cbij} by providing a bijection between the sets $A^{C}_k(n)$ and $M^{C}_{k}(n)$, $k \in \{0\}\cup [n]$. Before proceeding to the proof of  Theorem \ref{cbij}   we partition the sets  $A^{C}_k(n)$ and $M^{C}_{k}(n)$, $k \in \{0\}\cup [n]$, further. 
 
 Partition  $A^{C}_k(n)$, $k  \in \{0\}\cup [n]$, according to the $k$ distinct nonzero absolute values of the numbers appearing in the  sequence  and the number of times they appear. If $$\{|a_1|, |a_2|, \ldots, |a_n|\} \backslash \{0\}=\{c_1<c_2<\cdots<c_k\}$$ and $c_i$ appears $o_i$ times in $(|a_1|, |a_2|, \ldots, |a_n|)$, $ i \in [k]$, let 
${A^{C}_k}^{\c, \o}(n)$ equal
$$\Big\{(a_1, a_2, \ldots, a_n) \in A^{C}_k(n) \Big|  
 \{\{|a_1|, |a_2|, \ldots, |a_n|\}\}=\cup_{i=1}^{k}\cup_{j=1}^{o_i}\{\{c_i\}\} \cup_{i=1}^{n-\sum_{j=1}^k o_j} \{\{0\}\}\Big\},$$
 where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i\leq n$.
 
 For an antichain $\x \in M^{C}_k(n)$, let $(S_{\pi_{\x}}, g_{\pi_{\x}})$ be the pair of $k$-set and function corresponding to $\pi_{\x}$ under the bijection described in Theorem \ref{1}. Let  $$S_{\pi_{\x}}=\{c_1< \ldots< c_k\} \text{ and } o_i=g_{\pi_{\x}}(c_i), i \in [k].$$ Denote   $c(\x)=(c_1< \ldots< c_k)$ and  $o(\x)=(o_1, \ldots, o_k).$
 
  Partition the multiset $M^{C}_{k}(n)$, $k \in \{0\} \cup  [n]$, according to $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i\leq n$, as described above. Let 
$${M^{C}_{k}}^{\c, \o}(n)=\{\{\x \in M^{C}_{k}(n)|  
c(\x)=\c, o(\x)=\o \}\},$$
where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i\leq n$.


\begin{lemma} \label{cun}
The vectors $c(\x)=\c$ and $o(\x)=\o$, where  $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $k \in \{0\}\cup  [n]$, $o_i>0$, for $i\in [k]$, $\sum_{i=1}^k o_i\leq n$,  uniquely determine the antichain $\x$.
\end{lemma}
 
 \proof Lemma \ref{cun} follows readily since Theorem \ref{1} establishes a bijection.
 \qed
 
 
 \begin{theorem} \label{cco}
 $$|{A^{C}_k}^{\c, \o}(n)|=|{M^{C}_{k}}^{\c, \o}(n)|={n \choose {o_1, \ldots, o_k, n-\sum_{j=1}^k o_j}}2^{\sum_{j=1}^k o_j}, $$
 where $k \in \{0\} \cup [n]$, $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $o_i>0$, for $i\in [k]$, and $\sum_{i=1}^k o_i\leq n$.
 \end{theorem}
  
    \proof A bijective proof of the first equality   can be given using Theorem \ref{1} and the ideas of Theorem \ref{3}. The enumeration is in Lemma \ref{2}.  Note that arrangements do not enter any of the proofs.  \qed
    \medskip
  
  
 \noindent \textit{Proof of Theorem \ref{cbij}.} Straightforward corollary of Theorem \ref{cco}, since $$A^{C}_{k}(n)=\sum_{\c, \o} {A^{C}_{k}}^{\c, \o}(n)=\sum_{\c, \o} {M^{C}_{k}}^{\c, \o}(n)=M^{C}_{k}(n),$$  where $\c=(c_1< \ldots< c_k)$, $\o=(o_1, \ldots, o_k)$, $k \in [n]$, $o_i>0$, for $i\in [k]$, $\sum_{i=1}^k o_i\leq n$.
 \qed
 
 
 \begin{corollary}\label{cor:bb}
$$ \sum_{w \in \S_n^C} j_P(Q^{C}_w)=(2n+1)^{n}.$$
 \end{corollary}
 
 \proof Theorems \ref{cco} and \ref{cbij} extend to a bijection between $$\mathcal{M}^{C}(n)=\cup_{k=0}^n \cup_{\c, \o}{M^{C}_{k}}^{\c, \o}(n) \mbox{ and }\A^{C}(n)=\cup_{k=0}^n \cup_{\c, \o}{A^{C}_{k}}^{\c, \o}(n),$$ the cardinalities of which are  $ \sum_{w \in \S_n^C} j_P(Q^{C}_w)$ and $(2n+1)^{n}$, respectively.
\qed
 
 
 

\subsection*{Acknowledgement}
  I would like to thank Richard Stanley for his beautiful  lectures,  which served as an inspiration  for this paper. I would also like to thank Drew Armstrong for the many thoughtful suggestions and references he provided as well as the anonymous referee for his careful reading and many constructive suggestions.






\end{document}
